Author's response to reviews

Title: Three years of pulmonary rehabilitation: inhibit the decline in airflow obstruction, improve exercise endurance time, and body-mass index, in chronic obstructive pulmonary disease

Authors:

David Stav (dstav@asaf.health.gov.il)
Meir Raz (raz_m@maccabi.org.il)
Isaac Shpirer (i_shpirer@yahoo.com)

Version: 6 Date: 22 May 2009

Author's response to reviews: see over
Dear editor

Additional changes as suggested were made in the manuscript.

1. The titles are now identical
2. Each affiliation was written in full, as required.
3. All authors read and approved the final manuscript.
4. Tables - Tables were created using a Tables tool, and not were linked to files, figures, or generated with tabbed text. Tables were divided into rows and columns and were oriented in a portrait page layout.
5. Figure legends - Figure legend 1 graphics were written in text.

David Stav